Author’s response to reviews

Title: Glucose challenge increases circulating progenitor cells in Asian Indian male subjects with normal glucose tolerance which is compromised in subjects with pre-diabetes: A pilot study.

Authors:

Abel A Nathan (abelzone@gmail.com)  
Viswanathan Mohan (drmohans@vsnl.net)  
Subash S Babu (sbabu@mail.nih.gov)  
Soumi Bairagi (soumi.b2005@gmail.com)  
Madhulika Dixit (mdixit@iitm.ac.in)

Version: 4 Date: 29 November 2010

Author’s response to reviews: see over